**Author's response to reviews**

**Title:** Polymorphisms in regulatory regions of Cyclooxygenase-2 gene and breast cancer risk in Brazilians: a case-control study

**Authors:**

Diogo N Piranda (diogopiranda@yahoo.com.br)  
Juliana S Festa-Vasconcellos (juliana.s.festa@gmail.com)  
Laura M Amaral (murta.laura@gmail.com)  
Anke Bergmann (abergmann@inca.gov.br)  
Rosane Vianna-Jorge (farmaco@inca.gov.br)

**Version:** 14  **Date:** 4 November 2010

**Author's response to reviews:** see over
A few typographical errors were corrected in the text. Tables’ format was revised so that the size of letters and the alignments were fixed. In relation to the payment, it will be done by the Financial Bureau of the Brazilian National Cancer Institution, which will take a few days to process the fee.

We thank you for the careful revision of the manuscript.

Sincerely yours,

Rosane Vianna-Jorge, PharmD, PhD
Associate Professor
Institute of Biomedical Sciences – Federal University of Rio de Janeiro
Head of Research Group, Clinical Pharmacology and Pharmaceutical Assistance
Division of Pharmacology, Brazilian National Cancer Institute
farmaco@inca.gov.br
Phone: + 55 21 3233-1292 Fax : +55 21 3233-1340